

## ANSWERING REVIEWERS

September, 24<sup>th</sup>, 2014



Dear Editor,

We are pleased to hear that our manuscript has been reviewed and it can be considered for publication.

We are also very grateful to the referees since we believe that the changes we have introduced in response to his suggestions have improved the manuscript in the enclosed final form. Please find bellow all the modifications done.

We hope that after evaluating our revision, you will find that we have dealt successfully with all issues raised by the reviewer and that the manuscript is now acceptable for publication

Please find enclosed the edited manuscript in Word format (file name: 13988- final review.doc).

**Title: From minimal to maximal surgery in treatment of hepatocarcinoma: a review**

**Name of Journal:** *World Journal of Hepatology*

**ESPS Manuscript NO:** 13988

The manuscript has been improved according to the suggestions of reviewer.

1 - Name of the article was changed as per reviewer and Editorial Board (less than 12 words)

2- Suggestions of the reviewer

- a. The structure of the article was changed and items were added to the sections
- b. Bibliography was reviewed and added.

3- English polishing was performed by the senior authors.

Thank you again for publishing our manuscript in the *World Journal of Hepatology*. And hope that in the future I can be of any help.

Sincerely yours,

Marcos Vinicius Perini, MD, PhD  
Austin Hospital  
Department of Surgery - University of Melbourne  
Heidelberg - Australia  
E-mail: marcos.perini@usp.br